Author's response to reviews

Title: Differential association between metabolic syndrome and coronary artery disease evaluated with cardiac computed tomography according to the presence of diabetes in a symptomatic Korean population

Authors:

Ki-Bum Won (kbwon99@naver.com)
Hyuk-Jae Chang (hichang@yuhs.ac)
Gue-Ru Hong (grhong@yuhs.ac)
Jimin Sung (jmsung77@gmail.com)
Sanghoon Shin (NEPHILLA@yuhs.ac)
In-Jeong Cho (injeongcho@naver.com)
Chi-Young Shim (cysprs@yuhs.ac)
Young Jin Kim (dryj@yuhs.ac)
Byung-Wook Choi (bchoi@yuhs.ac)
Namsik Chung (namsikc@yuhs.ac)

Version: 5
Date: 1 April 2014

Author's response to reviews:

I corrected the wrong author details.